Author’s response to reviews

Title: Identification of priority areas for surveillance of cutaneous leishmaniasis using spatial analysis approaches in southeastern of Brazil

Authors:
Diogo Cardoso (diogo.tavares0@yahoo.com.br)
Dayane de Souza (dhay.cdss@gmail.com)
Vanessa de Castro (vanessanormandio@hotmail.com)
Stefan Geiger (stefan.geiger76@gmail.com)
David Barbosa (davidsoeiro@gmail.com)

Version: 2 Date: 07 Mar 2019

Author’s response to reviews:

INFD-D-18-00295R1

Identification of priority areas for surveillance of cutaneous leishmaniasis using spatial analysis approaches in southeastern of Brazil

Diogo Tavares Cardoso; Dayane Costa de Souza, M.D; Vanessa Normandio de Castro, M.D; Stefan Michael Geiger, M.D, PhD; David Soeiro Barbosa, M.D, PhD

BMC Infectious Diseases

Response upon re-submission: We thank all reviewers and the editors for their work and valuable comments. We have now prepared a new, clean version of our manuscript.
Technical Comments:

1. Please include all author e-mails on the title page.
Response: All author E-mails are now included on the title page

2. Please re-name the MATERIALS AND METHODS to Methods.
Response: The section has been re-named to Methods.

3. In Funding, please state whether or not the funding body (FAPEMIG) played any roles in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.
Response: DCT as a regular student in the Parasitology Post-Graduation Program received financial support from FAPEMIG. The financial support did not play any role in the design of the study or collection, analysis, and interpretation of data.

4. Please clarify the origin of the maps used in figures 1, 4, 5 and 6 in the figure legends. Could you clarify if you own the copyright for the map image or have permission to reproduce it (and that the copyright holder is aware that the figure will be published in an open access journal under a creative commons license)?
Response: The maps used in Figures 1, 4, 5, and 6 were downloaded as digitalized shapefiles from the public site of the Brazilian Institute of Geography and Statistics (IBGE; www.ibge.gov.br). Afterwards, these maps were modified with different colour codes by the authors, according to the epidemiological data.

5. Upon resubmission, please remove any tracked changes or highlighting and include only a single clean copy of the manuscript. Please ensure the document is in the final form for publication; please upload only files that are to be published and remove any files that should not form part of the final manuscript (e.g. covering letter, response to reviewers, etc). If you wish to respond to these revision requests, please insert the information in to the designated input box only
Response: A clean copy of the manuscript has been re-submitted.